Author's response to reviews

Title: Measuring nasal bacterial load and its association with otitis media

Authors:

Heidi C Smith-Vaughan (heidi@menzies.edu.au)
Roy Byun (roybyun@dental.wsahs.nsw.gov.au)
Mangala Nadkarni (mnadkarni@dental.wsahs.nsw.gov.au)
Nicholas A Jacques (njacques@dental.wsahs.nsw.gov.au)
Neil Hunter (nhunter@dental.wsahs.nsw.gov.au)
Stephen Halpin (stephen.halpin@menzies.edu.au)
Peter S Morris (peter.morris@menzies.edu.au)
Amanda J Leach (amanda@menzies.edu.au)

Version: 4 Date: 9 May 2006

Author's response to reviews:

Dear Sir/Madam

The requested formatting changes have been made to the manuscript.
1. A missing space has been placed in the first line of the abstract.
2. All authors are now listed in the bibliography.
3. Figures have been cropped.

Thank you for accepting the manuscript.

Yours sincerely,

Heidi Smith-Vaughan